ss13-Jun-03
Week 25: (16/06 to 20/06)

SM18 Operation SCHEDULE (Monday Morning) Meeting 

· Test Coordinator this week:   S.Sanfilippo (16-4641)

· MTA Technician on call:  (16-3602) E.Effinger.

1. Planning for this week : 
· TBA1:   3014  :  Test at 1.9 K.
· TBA2:   1016 Test at 1.9 K _end and warm up.
· TBC2 :  1026 :  Test at 1.9 K (1 run foreseen)
·  TBC1 : 1024  Waiting for removal: 2008 will come
2.  Problems from Operation Last week ( Olav+ Federico)
· Rotary switch : be gentle with it.

· Problem with DCCT during one quench (TBA2). Solved?

· Offset in TT821: solved? 

· Resistance too low across the magnet 5.7 Ohm: new IFS box

· TT821sp not OK and no spare.

· TBC2 : Anticryostat cables not connected and temperature OK?

· TBC2 : LF file wrong (temperature too low)

· AQA problem: LF file missing: problem with transfer file.

· Still problem in saving quench table files after changing PC.

· AFS problem in the console used for TBA2.

· MM problems : Check that the coil factor corresponds to the position of the switch in the rack.

· VME (communication) errors during a MM : Integrator problem.

· During the export of MM (cable coupling current) no confirmation.

· Remember please to fill test sequence, quench tables and check MM with display (fill the protocols to report errors)
3. Situation From Cryogenics:  J-P Lamboy
4. Other General Issues
